Author's response to reviews

Title: The prevalence of hyperuricemia in China: a meta-analysis

Authors:

Bin Liu (liubin409@163.com)
Tao Wang (wtwt128@yahoo.com.cn)
Ni Hai Zhao (deararyuemoon@hotmail.com)
Wei Wen Yue (yuewenwei@medmail.com.cn)
Peng Hua Yu (yuhuapeng1972@163.com)
Xia Cui Liu (liucui_xia@126.com)
Jie Yin (yinjie12345@sohu.com)
Yi Ru Jia (jrycardiology@sohu.com)
Wei Hong Nie (niehw1985@163.com)

Version: 7 Date: 28 September 2011

Author's response to reviews:

Author style was revised, initials was used in authors’ contributions, references 77, 78 and 79 were included in the reference list and table 4 was removed.